Author's response to reviews

Title: Predicting Streptococcal Pharyngitis in Adults in Primary Care: A Systematic Review of the Diagnostic Accuracy of Symptoms and Signs and Validation of the Centor Score

Authors:

Jolien Aalbers (j.aalbers@student.ru.nl)
Kirsty K O'Brien (kirstyobrien@rcsi.ie)
Wai-Sun Chan (wschan@rcsi.ie)
Gavin Falk (gavfalk@gmail.com)
Conor Teljeur (cteljeur@hiqa.ie)
Borislav D Dimitrov (borislavdimitrov@rcsi.ie)
Tom Fahey (tomfahey@rcsi.ie)

Version: 2 Date: 29 March 2011

Author's response to reviews:

Dear Sir/Madam,

No further revisions were requested in this version of the manuscript. This manuscript was originally submitted to BMC Family Practice and we were invited to publish in BMC Medicine after revisions were made.

Yours sincerely

Kirsty O'Brien